Author's response to reviews

Title: Household's Perception of Climate Change and Human Health Risks: A community perspective

Authors:

Md. Aminul Haque (aminuldp@gmail.com)
Shelby Suzanne Yamamoto (shelby.yamamoto@urz.uni-heidelberg.de)
Ahmad Azam Malik (Malik@uni-heidelberg.de)
Rainer Sauerborn (Rainer.Sauerborn@urz.uni-heidelberg.de)

Version: 2 Date: 26 August 2011

Author's response to reviews:

Just in the title of the article
perception of in place of perception on